%~MoulinÃ© par MaN_auto v.0.34.0 (b272df1a) 2025-06-03 20:26:15
\documentclass[CRMATH,Unicode,XML,biblatex]{cedram}

\TopicFR{Analyse harmonique}
\TopicEN{Harmonic analysis}

\addbibresource{CRMATH_Moritoh_20241101.bib}

\usepackage{mathtools}

\newcommand*{\dd}{\mathrm{d}}

%%%---------------------------------------------------------------------------

%%% DELIMITERS

%\usepackage{mathtools}
\DeclarePairedDelimiter{\parens}{\lparen}{\rparen}
\DeclarePairedDelimiter{\braces}{\{}{\}}
\DeclarePairedDelimiter{\abs}{\lvert}{\rvert}
\DeclarePairedDelimiter{\norm}{\lVert}{\rVert}
\DeclarePairedDelimiter{\bracks}{[}{]}
\DeclarePairedDelimiter{\bbracks}{\llbracket}{\rrbracket}
\DeclarePairedDelimiter{\angles}{\langle}{\rangle}
\DeclarePairedDelimiter{\aangles}{\langle\!\langle}{\rangle\!\rangle}
\DeclarePairedDelimiter{\pparens}{(\!(}{)\!)}

%%%---------------------------------------------------------------------------

\graphicspath{{./figures/}}

\newcommand*{\mk}{\mkern -1mu}
\newcommand*{\Mk}{\mkern -2mu}
\newcommand*{\mK}{\mkern 1mu}
\newcommand*{\MK}{\mkern 2mu}

\hypersetup{urlcolor=purple, linkcolor=blue, citecolor=red}

%%%---------------------------------------------------------------------------

%%% Changer la largeur des adresses

% Jouer avec \hsize :
\makeatletter
\def\@setafterauthor{%
  \vglue3mm%
%  \hspace*{0pt}%
\begingroup\hsize=12cm\advance\hsize\abstractmarginL\raggedright
\noindent
%\hspace*{\abstractmarginL}\begin{minipage}[t]{10cm}
   \leftskip\abstractmarginL
  \normalfont\Small
  \@afterauthor\par
\endgroup
\vskip2pt plus 3pt minus 1pt
}
\makeatother

%%%---------------------------------------------------------------------------

\title{A direct proof of the weighted P\'olya--Knopp inequality following Carleson's method}

\alttitle{Une preuve directe de l'in\'egalit\'e de P\'olya--Knopp
\`a poids par la m\'ethode de Carleson}

\author{\firstname{Emu} \lastname{Kondo}}
\address{Graduate School of Humanities and Sciences, Nara Women's University, Nara, Japan 630-8506}
\email{wae\_kondo@cc.nara-wu.ac.jp}

\author{\firstname{Shinya} \lastname{Moritoh}}
\address{Department of Mathematics, Nara Women's University, Nara, Japan 630-8506}
\email{moritoh@cc.nara-wu.ac.jp}

\author{\firstname{Yumi} \lastname{Tanaka}}
\address{Center for Diversity and Inclusion, Nara Women's University, Nara, Japan 630-8506}
\email{yumi\_mrth\_logx@hera.eonet.ne.jp}

\subjclass{26D15, 26A51}

\keywords{\kwd{Hardy's inequality} \kwd{P\'olya--Knopp's inequality} \kwd{weight} \kwd{doubling condition} \kwd{nonincreasing}}
\altkeywords{\kwd{InÃ©galitÃ© de Hardy} \kwd{inÃ©galitÃ© de PÃ³lya--Knopp} \kwd{poids} \kwd{doublement} \kwd{convexitÃ©}}

\begin{abstract}
The aim of the paper is to provide a direct proof of the weighted P\'olya--Knopp inequality. This inequality (which is a limiting case of the Ari\~no--Muckenhoupt inequalities), involving non-increasing functions, was initially established by Sbordone--Wik, who proved its validity under the necessary and sufficient condition that the weight satisfies an appropriate doubling condition. Our main contribution is to use Carleson's approach to Carleman's inequality in conjunction with Hardy's lemma and Sbordone--Wik's doubling condition, in order to obtain the weighted P\'olya--Knopp inequality.
\end{abstract}

\begin{altabstract}
L'objectif de cet article est de fournir une preuve directe de l'in\'egalit\'e de P\'olya--Knopp \`a poids. Cette in\'egalit\'e (qui est un cas limite des in\'egalit\'es de Ari\~no--Muckenhoupt), portant sur des fonctions d\'ecroissantes, a \'et\'e montr\'ee par Sbordone--Wik, qui ont montr\'e qu'elle est valide si elle seulement si le poids satisfait une condition adapt\'ee de doublement. Notre contribution principale est d'utiliser la m\'ethode de Carleson pour la preuve de l'in\'egalit\'e de Carleman, en conjonction avec le lemme de Hardy et la condition de doublement de Sbordone--Wik, afin de prouver directement l'in\'egalit\'e de P\'olya--Knopp \`a poids.
\end{altabstract}

\begin{document}

\maketitle

\section{Introduction}

Let the Ari\~no--Muckenhoupt class $B_p\,\,(1 \le p <\infty)$ be defined as the set of all nonnegative functions $W$ for which a constant $B$ exists such that
\[
\int_r^\infty \parens[\Bigg]{\frac{r}{x} }^p W(x) \, \dd x
\le B \int_0^r W(x) \, \dd x \quad \text{for all $r>0$},
\]
and let the class $B_\infty$ be the union of all such $B_p$.

\goodbreak

Let $1 \le p <\infty$. Then, the Ari\~no--Muckenhoupt theorem~\cite[Theorem~1.7]{AM} asserts the following.
\begin{enonce*}{Ari\~no--Muckenhoupt theorem (\cite[Theorem~1.7]{AM})}
The weighted Hardy inequality
\[
\int_0^\infty
\parens[\Bigg]{\frac{1}{x} \int_0^x f(t)^{1/p} \, \dd t }^p W(x) \, \dd x
\le C \int_0^\infty f(x) \,W(x) \, \dd x
\]
holds for all nonnegative, \emph{nonincreasing} functions $f$ on $[0,\infty)$ if and only if $W$ belongs to the class~$B_p$.
\end{enonce*}

By Riesz~\cite{R}, $\parens[\big]{\int_0^x \,f(t)^{1/p} \, \dd t/x}^p$ decreases and tends to $\exp \parens[\big]{\int_0^x \log \, f(t) \, \dd t/x}$ as $p$ increases to $\infty$. Therefore, as a limiting case of this theorem, it is natural to try to characterize the class $B_{\infty}$ such that the following holds, where we use the notation $Gf(x)=\exp\parens[\big]{\int_0^x\log f(t) \dd t/x}$: the weighted P\'olya--Knopp inequality
\begin{equation}\label{eq2}
\int_0^\infty Gf(x)W(x) \, \dd x
\le C \int_0^\infty f(x) W(x) \, \dd x
\end{equation}
holds for all positive, nonincreasing functions $f$ on $[0,\infty)$ if and only if $W \in B_\infty$.

In Sbordone--Wik~\cite{SW}, the class $B_\infty$ is defined using the doubling condition~\eqref{cond_D_a} below. Moreover, it is proved in~\cite[Theorem~5]{SW} that the class $B_\infty$ is the union of all $B_p\,\,(1\le p<\infty)$, and the inequality~\eqref{eq2} is proved in~\cite[Theorem~6]{SW}.

The aim of this paper is to provide a direct proof of the inequality~\eqref{eq2} by making full use of Carleson's approach to Carleman's inequality.

Following Carleson~\cite{C}, we express a nonincreasing function $f(x)$ on $(0,\infty)$ by using a nonincreasing function $h(x)$ as $ f(x)= e^{h (x)}$, where $h \colon (0, \infty) \to {\mathbb R} \cup \{\pm\infty\}$. Then $Gf(x)= e^{H(x)}$ and the inequality~\eqref{eq2} can be written as
\begin{equation}\label{eq11}
\int_0^\infty e^{H(x)} W(x) \, \dd x
\le C \int_0^\infty e^{h(x)} W(x) \, \dd x,
\end{equation}
where
\[
H(x)\coloneqq \frac{1}{x} \int_0^x h(t) \, \dd t, \quad x\in(0, \infty).
\]
The inequality where $W(x)=x^{\,p} \,(-1<p <\infty)$ and $C=e^{p+1}$ is treated in Carleson~\cite{C}. The case where $p=0$ is known as the P\'olya--Knopp inequality~\cite{K}. See also Carleman~\cite{Carm}.

Our paper is organized as follows. In Section~\ref{section_2}, we restate the theorem to be considered and then give a direct proof of the inequality~\eqref{eq11} following Carleson's method. A few comments are mentioned in Section~\ref{section_3}.

\section{Theorem and a direct proof of the weighted PÃ³lya--Knopp inequality}\label{section_2}

For the convenience of the reader, we restate the result to be proved.

\begin{theorem}
Let the weight function $W$ belong to the class $B_\infty$. Then the weighted P\'olya--Knopp inequality~\eqref{eq11} holds for all nonincreasing function $h \colon (0, \infty) \to {\mathbb R} \cup \{\pm\infty\}$, where $H(x)= (1/x) \int_0^x h(t) \, \dd t$.
\end{theorem}

\begin{proof}
For a nonnegative function $W(x)$, suppose that it satisfies the $B_{\infty}$ condition, or equivalently, the following doubling condition~\eqref{cond_D_a} for some $a>1$:
\begin{equation}\label{cond_D_a}
\int_0^{ar} W(x) \,\dd x \le C_a \int_0^rW(x)\,\dd x \quad \text{for all $r>0$}.
\tag{$\mathrm{D_a}$}
\end{equation}
For this equivalence, see Sbordone--Wik~\cite[Theorem~5]{SW}. The inequality~\eqref{eq11} is now to be proved. Because $h$ is nonincreasing, we have $axH(ax) \le xH(x) + (ax-x)h(x)$ (even in the case where $h(x) = -\infty$ and thus this inequality becomes $-\infty \le -\infty$). We use this inequality and H\"older's inequality to obtain, for any $A>0$,
\begin{equation}\label{eq100}
\begin{split}
\int_0^{A} e^{H(x)} W(x) \, \dd x 
& = \int_0^{A/a} e^{H(ax)} W(ax)  a \, \dd x \\
& \le \int_0^{A} e^{H(x)/a}\,e^{(a-1)h(x)/a}W(ax) a \, \dd x \\
& \le \parens[\Bigg]{\int_0^{A} e^{H(x)} W(ax)  a \, \dd x}^{1/a}
\parens[\Bigg]{\int_0^{A} e^{h(x)} W(ax)  a \, \dd x}^{(a-1)/a}.
\end{split}
\end{equation}
We next note that the functions $e^{H(x)}$ and $e^{h (x)}$ are nonincreasing, and that the condition~\eqref{cond_D_a} can be written as follows:
\[
\int_0^r W(ax) a \, \dd x \le C_a \int_0^r W(x)\,\dd x \quad \text{for all $r>0$}.
\]
We now recall Hardy's lemma according to~\cite[Proposition~3.6 in Chapter~2]{BS}.
\begin{enonce*}{Hardy's lemma (\cite[Proposition~3.6 in Chapter~2]{BS})}
Let $f_1$ and $f_2$ be nonnegative functions on $[0,\infty)$, and suppose
\[
\int_0^{r}f_1(x)\, \dd x \le \int_0^{r}f_2(x)\,\dd x \quad \text{for all $r>0$}.
\]
Let $g$ be any nonnegative nonincreasing function on $[0,\infty)$. Then,
\[
\int_0^{A}f_1(x)g(x)\,\dd x \le \int_0^{A}f_2(x)g(x)\,\dd x \quad \text{for all $A>0$}.
\]
\end{enonce*}
In our case, Hardy's lemma yields
\begin{equation}\label{eq101}
\int_0^{A} e^{H(x)} W(ax)  a \, \dd x
\le C_a \int_0^{A} e^{H(x)} W(x) \, \dd x,
\end{equation}
and
\begin{equation}\label{eq102}
\int_0^{A} e^{h(x)} W(ax)  a \, \dd x
\le C_a \int_0^{A} e^{h(x)} W(x) \, \dd x.
\end{equation}
Therefore, from~\eqref{eq100}, \eqref{eq101} and~\eqref{eq102}, we obtain
\begin{equation}\label{eq103}
\int_0^A e^{H(x)} W(x) \, \dd x
\le C_a \parens[\Bigg]{\int_0^A e^{H(x)} W(x) \, \dd x}^{1/a}
\parens[\Bigg]{\int_0^A e^{h(x)} W(x) \, \dd x}^{(a-1)/a}.
\end{equation}
As it is assumed in Ari\~no--Muckenhoupt~\cite[p.~730]{AM}, we can also assume that $h$ is constant on $[0, d]$ for some $d > 0$. We can add this restriction on $h$ without loss of generality by use of the monotone convergence theorem. Then $H$ is also constant on $[0, d]$ and we have
\[
\int_0^A e^{H(x)} W(x) \, \dd x < \infty.
\]
By rearranging the inequality~\eqref{eq103} using this finite-valued term, we obtain
\[
\int_0^A e^{H(x)} W(x) \, \dd x
\le C_a^{a/(a-1)} \int_0^A e^{h(x)} W(x) \, \dd x.
\]
Letting $A$ tend to $\infty$, we obtain the desired inequality~\eqref{eq11} with the constant $C=C_a^{a/(a-1)}$.
\end{proof}

\begin{remark}
In Carleson~\cite{C}, the power function $W(x)=x^p~(-1<p<\infty)$ is used as a weight. Then, the doubling constant is $C_a=a^{p+1}$, and we have that $\inf_{a>1}C_a^{a/(a-1)}=\lim_{a\to1}C_a^{a/(a-1)}= e^{p+1}$, which is known to be the best constant for the power-weighted P\'olya--Knopp inequality.
\end{remark}

\section{Comments}\label{section_3}

\begin{enonce}[remark]{Comment}
For the use of Carleson's approach in the related direction of the weighted Hardy inequality, see Kwon~\cite{Kw}.
\end{enonce}

\begin{enonce}[remark]{Comment}
Let $1\le p<\infty$. Then, the Hardy--Littlewood maximal operator is bounded on the classical Lorentz space $\Lambda_p(W)$ if and only if $W$ satisfies the $B_p$ condition. See Ari\~no--Muckenhoupt~\cite{AM}. The question that remains is what is the limiting case of this equivalence when $p\to\infty$.
\end{enonce}

\section*{Acknowledgments}

The authors would like to express their sincere gratitude to an anonymous referee for their very valuable comments on the representation of nonincreasing functions and on the treatment of infinities that may appear in the inequalities. They would also like to express their sincere gratitude to Professors Hitoshi Ishii (Waseda University) and Eiichi Nakai (Ibaraki University) for valuable suggestions and encouragement.

\section*{Declaration of interests}

The authors do not work for, advise, own shares in, or receive funds from any organization that could benefit from this article, and have declared no affiliations other than their research organizations.

\printbibliography

\end{document}